Author’s response to reviews

Title: Modifiable risk factors for falling and their associations with visiting the Accident & Emergency Department after a fall

Authors:

Alice C. Scheffer (a.c.scheffer@amc.nl)
Pieter Boele van Hensbroek (p.boelevanhensbroek@amc.nl)
Nynke van Dijk (n.vandijk@amc.nl)
Jan S.K. Luitse (j.s.luitse@amc.nl)
Johannes C. Goslings (j.c.goslings@amc.nl)
René H. Luigies (r.luigies@eusamed.eu)
Sophia E. de Rooij (s.e.derooij@amc.nl)

Version: 2 Date: 4 May 2012

Author’s response to reviews: see over
Revision article

BMC ID 6866591797170580 “Modifiable risk factors for falling and their associations with visiting the Accident & Emergency Department after a fall”.

4th May 2012

Dear Madam/ Sir,

Hereby, I sent you a revision of the above mentioned article. Below you will find the comments, the authors’ answers and the changes made in the manuscript.

On behalf of the other authors,

Yours sincerely,

Alice Scheffer, RN, PhD
Academic Medical Center
Department of Internal and Geriatric Medicine
F4-108
PO Box 22700
1100 DE Amsterdam, the Netherlands
Tel: + 31 20 5665991; fax: + 31 20 5669325
Email: a.c.scheffer@amc.uva.nl

COMMENTS:

Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

We have changed the text as suggested by the reviewer and added an ethics statement to the Methods section.